Author’s response to reviews

Title: Effect of Ketofol versus propofol as an induction agent on ease of laryngeal mask airway insertion conditions and hemodynamic stability in pediatrics: An observational prospective cohort study.

Authors:

Bacha Aberra (bcabera11@gmail.com)
Adugna Aregawi (adugenet@gmail.com)
Girmay Teklay (girmeat@gmail.com)
Hagos Tasew (tasewh2@gmail.com)

Version: 2 Date: 06 Jun 2018

Author’s response to reviews:

Title: Effect of Ketofol versus propofol as an induction agent on ease of laryngeal mask airway insertion conditions and hemodynamic stability in pediatrics in Menelik-II Hospital, Addis Ababa, Ethiopia: Observational prospective cohort study.

Thank you for the constructive comments

We made revision on the main manuscript. In our study we revised the methodology of the study in a clear way. It is an observational prospective cohort study. And you can see it in the methodology section line number 117-118

The whole manuscript is also revised according to the STROBE check list. you can see throughout the manuscript.

We also made revision on the outcomes of the study to be clear. You can see it in methodology section line number 131-134 and 173-205.

The sample size of the study was calculated using the epi info ver.7.1.1 and we take the final sample size from Fleiss in the StatCalc. You can see it in methodology section line number 156-163

Finally we tried to modify the language errors throughout the manuscript.